Author’s response to reviews

Title: General health status in Iranian diabetic patients assessed by Short-Form-36 questionnaire: a systematic review and meta-analysis

Authors:
Masoud Behzadifar (masoudbehzadifar@gmail.com)
Rahim Sohrabi (sohrabi2003@gmail.com)
Roghayeh Mohammadibakhsh (r.mohammadib@yahoo.com)
Morteza Salemi (morsal59@gmail.com)
Sharare Taheri Moghadam (sh.mtaheri@yahoo.com)
Masood Taheri Mirghaed (masood.mirghaed@gmail.com)
Meysam Behzadifar (m_behzadifar67@yahoo.com)
Hamid Reza Baradaran (saaz2000@yahoo.com)
Nicola Luigi Bragazzi (robertobragazzi@gmail.com)

Version: 1 Date: 23 Jan 2018

Author’s response to reviews:

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are many issues with grammar, wording, spelling, and/or punctuation that need to be addressed.

We revised with native English speaking colleague.

Please note that use of a professional language service is not a guarantee of acceptance for publication.

2. We note that your paper has been submitted as a systematic review and meta-analysis; however you refer to the use of STROBE and the collection of primary data. Please provide further clarification regarding your study design?
3. In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/submissions/editorial-policies#standards+of+reporting), could you please ensure your manuscript reporting adheres to PRISMA guidelines (http://www.prisma-statement.org/) for reporting systematic reviews. This is so your methodology can be fully evaluated and utilised. Can you please include a completed PRISMA checklist as an additional file when submitting your revised manuscript.

Please complete the checklist in full by inserting the page number/paragraph and section of your manuscript which reports the information that meets the criteria of the checklist. For example “Methods, paragraph 2”. If a criterion is not applicable for your particular manuscript/study, we can accept “N/A”.

We doing and added in PRISMA checklist.

Please note that checklists completed incorrectly will be returned for revision as we cannot progress your manuscript to peer review until the checklist has been completed.

4. Please ensure that your manuscript is formatted as per the submission guidelines at https://bmcendoerdisord.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

5. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors
6. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

We dine and revised.

7. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

We done and revised.

8. At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.

Ok. We revised and not include track changes or highlighting.